Author’s response to reviews

Title: Novel PYGL mutations in Chinese children leading to glycogen storage disease type VI: two case reports

Authors:

Xiaomei Luo (luoxiaomei@xinhuamed.com.cn)
Jiacheng Hu (hujiacheng@sjtu.edu.cn)
Xueren Gao (gaoxueren@xinhuamed.com.cn)
Yanjie Fan (fanyanjie@xinhuamed.com.cn)
Yu Sun (sunyu@xinhuamed.com.cn)
Xuefan Gu (guxuefan@xinhuamed.com.cn)
Wenjuan Qiu (qiuwenjuan@xinhuamed.com.cn)

Version: 4 Date: 13 Mar 2020

Author’s response to reviews:

We are grateful for the comments of the editor. We have addressed all the comments. The revised manuscript of clean version has been submitted.

1. Consent for publication

A: We thank the reviewer for the comment. This section has been clarified (page 9).

2. Availability of Data and Materials

A: We thank the reviewer for the comment. The phenotypes and mutations of the two GSD VI patients were submitted to Leiden Open Variation Database under accession number #00295511 and #00295495. This section has been edited (page 10).

3. Authors’ contributions

A: We thank the reviewer for the comment. The author JH participated in revising the manuscript and did a part of the clinical work. This section has been modified (page 10).
4. Abbreviations

A: We thank the reviewer for the comment. This section has been checked (page 9).

5. Remove attachments

A: We thank the reviewer for the comment. This section has been modified (page 12).